Author's response to reviews

Title: Knowledge and attitudes of primary care physicians in the management of patients at risk for cardiovascular events

Authors:

Hamidreza Doroodchi (hamid@ceoutcomes.com)
Maziar Abdolrasulnia (mazi@ceoutcomes.com)
Jill A Foster (jill.foster@ceoutcomes.com)
Elyse Foster (foster@medicine.ucsf.edu)
Mintu P Turakhia (mintu.turakhia@ucsf.edu)
Kimberly A Skelding (kaskelding@geisinger.edu)
Kiran Sagar (ksagar@attglobal.net)
Linda L Casebeer (linda.casebeer@ceoutcomes.com)

Version: 5 Date: 3 July 2008

Author's response to reviews:

"*These authors contributed equally to this work" statement was changed to "All authors contributed equally to this work"